Author's response to reviews

Title: Clinicopathological and prognostic significance of HER-2/neu and VEGF expression in colorectal carcinomas

Authors:

Li Qingguo (oncosurgeonli@sohu.com)
Daorong Wang (drdrwyz@hotmail.com)
Jing Li (lijyz@sohu.com)
Ping Chen (chen86ky@126.com)

Version: 2 Date: 22 September 2010

Author's response to reviews: see over
Dear Editor:

For my mistake, there are some grammar and syntax. Enclosed please find our manuscript entitled “Clinicopathological and prognostic significance of HER-2/neu and VEGF expression in colorectal carcinomas” by Qingguo Li, Daorong Wang, Jing Li, Ping Chen as original contribution for publication in BMC Cancer. All above-mentioned authors have contributed to the work, Professor Chen designed the study, the data were collected by Qingguo Li, the immunohistochemistry were done by Daorong Wang, Jing Li and the results was analyzed by Jing Li, Ping Chen, and the article was written by Qingguo Li and made minor alterations by Daorong Wang. All authors agree on the presented findings and familiar with the contents of the final paper and take responsibilities for the authenticity of data used in the paper. We confirmed that the research was approved by our hospital ethics committee and complied with the Helsinki Declaration.

This manuscript is original and has not been previously published, nor has it been simultaneously submitted to any other journal. This paper has no conflicts of interest.

We are looking forward to having your peer-review of this manuscript.

Thank you very much!

Sincerely yours, Qingguo Li